Author's response to reviews

Title: Screening for inborn errors of metabolism in high risk children: a three-year study in Zhejiang Province, China

Authors:

Xinwen Huang Dr (xinwenhuang@126.com)
Lili Yang Dr. (lilyang.001@gmail.com)
Fan Tong Dr. (451446720@qq.com)
Rulai Yang Dr. (nesily_yang@yahoo.com.cn)
Zhengyan Zhao Prof. (zhaozy@zju.edu.cn)

Version: 2 Date: 23 July 2011

Author's response to reviews:

Dear Editors,

About three decades have passed since the first newborn screening program (for PKU) was introduced into China. There are still many problems in newborn screening in China. As for the screening for congenital hypothyroidism (CH), low screening rate throughout china, no united cut-off value for screening and so on.

This paper has not been submitted or is under consideration for publication in another journal. The study complies with current ethical consideration. We also confirm that all the listed authors have participated actively in the study, and have seen and approved the submitted manuscript. The authors do not have any possible conflicts of interest.

Hope the paper can meet the high standard of the journal and we are looking forward to your positive response.

Yours sincerely,

Zhengyan Zhao, Professor of Pediatrics

Children's Hospital, Zhejiang University School of Medicine

Hangzhou 310003, China

Email: zhaozy@zju.edu.cn